Author's response to reviews

**Title:** Suggestive linkage detected for blood pressure related traits on 2q and 22q in the population on the Samoan islands

**Authors:**

Karolina Åberg (kaaberg@vcu.edu)
Feng Dai (fed1@pitt.edu)
Satupaitea Viali (sviali@yahoo.com)
John Tuitele (jtuitele03@yahoo.com)
Guangyun Sun (sungn@email.uc.edu)
Subba R Indugula (subba.indugula@uc.edu)
Ranjan Deka (dekar@ucmail.uc.edu)
Daniel E Weeks (weeks@pitt.edu)
Stephen T McGarvey (Stephen_McGarvey@brown.edu)

**Version:** 3 **Date:** 14 October 2009

**Author's response to reviews:** see over
Dear Editor in Chief,

Thank you for accepting our revised manuscript “Suggestive linkage detected for blood pressure related traits on 2q and 22q in the population on the Samoan islands”. We made the major and minor revisions and checked the manuscript. It is ready for publication. Please let me know if additional changes are needed.

For the authors,

Stephen T. McGarvey, PhD, MPH
Professor of Community Health
Brown University
Email: Stephen_McGarvey@brown.edu